Author’s response to reviews

Title: A Qualitative Assessment of Practitioner Perspectives Post-Introduction of the First Continuous Professional Competence (CPC) Guidelines for Emergency Medical Technicians in Ireland.

Authors:

Shane Knox (shaneknox7@gmail.com)
Suzanne Dunne (suzanne.dunne@ul.ie)
Walter Cullen (walter.cullen@ul.ie)
Colum Dunne (colum.dunne@ul.ie)

Version: 2 Date: 5 February 2015

Author’s response to reviews:

Dear Editor

I have revised the first manuscript submitted. The revisions are:

Line 179: I have included a link to the tables at the end of the document. Line 179 now states: (supplementary information for Question 3 is presented in Table 2)

Line 405: this now specifies the additional/added files: Table 1 and Table 2 (this line was not included in the original submission)

There has been no additional information added or any changes made to the manuscript other than these two changes.

Thank you.